Author's response to reviews

Title: Efficacious, Effective, and Embedded Interventions: Implementation Research in Infectious Disease Control

Authors:

Pascale Allotey (pascale.allotey@brunel.ac.uk)
Daniel D Reidpath (daniel.reidpath@brunel.ac.uk)
Hashim Ghalib (ghalibh@who.int)
Franco Pagnoni (pagnonif@who.int)
William C Skelly (chris.skelly@brunel.ac.uk)

Version: 3 Date: 12 September 2008

Author's response to reviews:

The sentence at the end of the paragraph on page 7 has been inserted.
Affiliations have been written in full.
The abstract has been reformatted in structured form.
References are consecutively numbered and in square brackets.
Footnotes have been integrated into text.
Figure placement and additional files text has been removed